\documentclass[12pt]{article}
\input{macros-course}

\showsolutionsfalse

\usepackage{hyperref}

\begin{document}
\prprob{2}{September 10, 2001}{Wednesday September 12}

%% Note: had to do this because the handout command doesn't wrap the
%% due info correctly (runs off the page!!!) needs to be fixed (rad)

\textbf{Reading:} Week 2 Lecture Notes.  Rosen \S 3.2--3.  Velleman \S
6.1, examples 6.2.3-4, \S6.3--4.

\vspace{.1in}

\textbf{By 9am on the due date, send email} to
\texttt{6042-probs@theory.lcs.mit.edu} indicating in at most three
sentences one or more of the following: the passage in the
reading---\emph{including its page number}---you 

\begin{itemize}

\item found most difficult,

\item found most surprising,

\item would like to have discussed in the next Lecture.

\end{itemize}

\input{prprob2-probs.tex}

\end{document}
